Author's response to reviews

Title: Noninvasive monitoring of myocardial function in a peritoneal metastasis rat model after surgical and cytostatic therapy: assessment with tissue Doppler and non-Doppler 2D strain echocardiography

Authors:

Jens Hartmann (jens.hartmann@charite.de)
Fabian Knebel (fabian.knebel@charite.de)
Stephan Eddicks (stephan.eddicks@charite.de)
Mark Beling (mark.beling@charite.de)
Andrea Grohmann (andrea.grohmann@charite.de)
Alexander Panda (alexander.panda@yale.edu)
Christoph A Jacobi (christoph.jacobi@charite.de)
Joachim M Muller (jmueller@charite.de)
Klaus-Dieter Wernecke (kdwernecke@sostana.com)
Gert Baumann (gert.baumann@charite.de)
Adrian C Borges (adrian.borges@charite.de)

Version: 4 Date: 3 July 2007

Author's response to reviews:

Dear Editor,

we have changed the format of the manuscript according to the list of the editorial team.

We hope that the text is now ready for publication.

Yours sincerely,

Fabian Knebel, Adrian C. Borges